Author’s response to reviews

Title: Value of signs, symptoms and plasma heart-type fatty acid-binding protein (H-FABP) in evaluating patients presenting with symptoms possibly matching acute coronary syndrome: background and methods of a diagnostic study in primary care

Authors:

Robert TA Willemsen (robert.willemsen@maastrichtuniversity.nl)
Frank Buntinx (Frank.Buntinx@med.kuleuven.be)
Bjorn Winkens (bjorn.winkens@maastrichtuniversity.nl)
Jan F Glatz (glatz@maastrichtuniversity.nl)
Geert Jan Dinant (geertjan.dinant@maastrichtuniversity.nl)

Version: 2
Date: 19 March 2014

Author’s response to reviews:

In answer to your mail I have added the following:

1) In the revised manuscript, I have added: the date of trial registration in the abstract, personal financial competing interests and an acknowledgements section.

2) Concerning study status and related articles I would like to inform you that we are now recruiting the first patients and our study will continue until appr. March 2015. So far, we have not published or offered for publication any work containing data of the study as described in this protocol.